Author’s response to reviews

Title: Primary myelolipoma presenting as polyp in the nasal cavity - A case report and review of literature

Authors:

Smiley Annie George (annsmiley78@gmail.com)
Marie Therese Manipadam (mtm2005@cmc.ac.in)
Regi Thomas (reganupthomas@cmc.ac.in)

Version: 2 Date: 18 January 2012

Author’s response to reviews:

Dear editor
Good day. I am hereby submitting the revised manuscript. The following changes have been made.
1. Title change
2. Ethnicity of the patient included.
3. Consent, competing interest and contribution sections moved before the reference list
4. Figure title and legend section added after the reference list.
5. Spelling corrected-common

Hoping that the changes requested are correctly addressed. Awaiting a favorable response.

Regards
Smiley Annie George